Author's response to reviews

Title: Alterations of Serum Biomarkers Associated with Lung Ventilation Function Impairment in Coal Workers: A Cross-sectional Study

Authors:

Jimin Zou (jinminzou@126.com)
Xianming du Prel Carroll (mingming30@yahoo.com)
Xianhong Liang (hh_liang@hotmail.com)
Dongmei Wang (wdm7252196@sina.com)
Chao Li (li_chao0296@sina.com.cn)
Baojun Yuan (ybj25999@163.com)
Sandra Leeper-Woodford (leeper_sk@mercer.edu)

Version: 7 Date: 24 September 2011

Author's response to reviews: see over
Author's response to reviews

Title: Alterations of Serum Biomarkers Associated with Lung Ventilation Function Impairment in Coal Workers: A Cross-sectional Study

Authors:
- Jimin Zou (jinminzou@126.com)
- Xianming du Prel Carroll (mingming30@yahoo.com)
- Xianhong Liang (hh_liang@hotmail.com)
- Dongmei Wang (wdm7252196@sina.com)
- Chao Li (li_chao0296@sina.com.cn)
- Baojun Yuan (ybj25999@163.com)
- Sandra Leeper-Woodford (leeper_sk@mercer.edu)

Version: 3 Date: September 16, 2011
Author's response to reviews: see over
Response to reviewer’s report, reviewer Ilker Ates

General

QUESTION

We are very grateful to your suggestions. We have made necessary changes in the Methods section.

Minor Essential Revisions

Methods:

Subjects:

1) We have moved this sentence to the bottom of the section.
2) Corrected.

Discussion:

1) Corrected.
2) We have re-arranged the sentence accordingly.
Response to reviewer’s report, reviewer Antonio Mutti

We are thankful for your comment and your recommendation for publication